Author's response to reviews

Title: Epidural lipomatosis and congenital small spinal canal in spinal anesthesia: a case report

Authors:

Per Flisberg (per.flisberg@skane.se)
Owain Thomas (owain.thomas@skane.se)
Bo Geijer (bo.geijer@lthalland.se)
Prof Ulf Schott (ulf.schott@skane.se)

Version: 10 Date: 27 April 2009

Author's response to reviews: see over
We hereby resubmit a new version of our paper aligned to the reviewers wishes. Epidural fat is given in the text in mm, indices are better described and fat grade and dural sac areas are also given in the Legends to figures. Reference list is aligned to your wishes with the two new references included and others withdrawn to keep the total number at 15.

Yours Sincerely

Ulf Schott
MD, PhD

University of Lund